Author's response to reviews

Title: Percutaneous Closure of Large VSD Using a Home-Made Fenestrated Atrial Septal Occluder in 18-Year-Old with Pulmonary Hypertension

Authors:

Hacer Kamali 1 (hacerdr@hotmail.com)
Turkay Saritas 1 (turkaysaritas@yahoo.com)
Abdullah Erdem 1 (drabdullaherdem@hotmail.com)
Celal Akdeniz 2 (celalakdeniz@yahoo.com)
Volkan Tuzcu 2 (vtuzcu@gmail.com)

Version: 2, Date: 25 March 2014

Author's response to reviews:

Dear Editor,

In accordance with your suggests, the new changes is as follows.

1. An acknowledgement has been added.
2. Line and page numbers have been added.
3. Reference citations have been arranged.
4. The table has been included in the main text in the table section following the references.

Sincerely,

Dr. Turkay SARITAS